Author’s response to reviews

Title: Striving to promote male involvement in maternal health care in rural and urban settings in Malawi - a qualitative study

Authors:

Lucy I Kululanga (lucykululanga@yahoo.co.uk)
Johanne Sundby (johanne.sundby@medisin.uio.no)
Address Malata (addressmalata@kcn.unima.mw)
Ellen Chirwa (embweza@kcn.unima.mw)

Version: 3 Date: 27 November 2011

Author's response to reviews: see over
Dear Editor

I am writing on behalf of my co-authors to thank you for accepting our manuscript to be published in your journal. We have made the following changes to the manuscript:

1. Author list – the list of authors in the manuscript have been written as in the submission system: Lucy I. Kululanga $^{1,2*}$, Johanne Sundby $^1$, Address Malata $^3$ and Ellen Chirwa $^3$

2. Affiliations – each institution have been included as a separate affiliation with its own number:

   $^1$Department of International Health, Section for Health and Society, Faculty of Medicine, University of Oslo, P.O. Box 1130, Blindern, Oslo 0318, Norway.

   $^2$Department of Community and Mental Health, Kamuzu College of Nursing, University of Malawi, P.O. Box 415 Blantyre, Malawi.

   $^3$Department of Maternal and Child Health, Kamuzu College of Nursing, University of Malawi, Private Bag 1 Lilongwe, Malawi.

3. E-mails – authors e-mail addresses have been included on the cover page:

   E-mails: Lucy I. Kululanga lucykululanga@yahoo.co.uk – Johanne Sundby johanne.sundby@medisin.uio.no - Address Malata addressmalata@kcn.unima.mw

   – Ellen Chirwa embweza@kcn.unima.mw

4. Authors contributions – we confirm that all authors read and approved the final manuscript. The section has been corrected.

5. The manuscript title has been written in sentence case.

6. Horizontal lines which were present in the manuscript have been removed.

7. Abbreviation section has been presented in a paragraph form.

8. Typographical and content errors have been corrected.
Thank you

Lucy Kululanga.